Author’s response to reviews

Title: Mobile phones are a viable option for surveying young Australian women: a comparison of two telephone survey methods

Authors:

Bette Liu (bliu@kirby.unsw.edu.au)
Julia ML Brotherton (jbrother@vcs.org.au)
David Shellard (david@hvrf.com.au)
Basil Donovan (bdonovan@kirby.unsw.edu.au)
Marion Saville (msaville@vcs.org.au)
John M Kaldor (jkaldor@unsw.edu.au)

Version: 6 Date: 24 November 2011

Author’s response to reviews:

Thank you for the proof. I have checked the document and made the following changes in the manuscript:

1. P8 Changed to read Australian born
2. Acknowledgement-corrected Australian National Health and Medical Research Council
3. Table 2: a few p-values transposed across incorrectly have now been corrected.

Kind Regards,

Bette